Author's response to reviews

Title: PCR melting profile (PCR MP) - a new tool for differentiation of Candida albicans strains

Authors:

Beata Krawczyk (beatak@chem.pg.gda.pl)
Justyna Leibner-Ciszak (justyna.leibner@wp.pl)
Anna Mielech (annamielech@gmail.com)
Magdalena Nowak (magdanowak25@wp.pl)
Józef Kur (kur@pg.gda.pl)

Version: 4 Date: 30 September 2009

Author's response to reviews:

Dear Dr. Sabina Alam
Scientific Editor
BMC series journals

Thank you very much for your letter and the referee letter on our resubmission (Ms: 1116649159278776). Reviewer’s report was without any remark. Due to the retrospective and mainly methodological study approval of the local ethics committee was not required according to the guidelines of our hospitals. The isolates used in this investigation were collected for routine diagnostic procedures by microbiological laboratories at the public hospital in Gdańsk and Szczecin and then were provided to our laboratory. We added this information in methods section.

Sincerely yours,

Beata Krawczyk